Author's response to reviews


Authors:

Fatma N Al-Maskari (fatma.am@uaeu.ac.ae)
Mohammed El-Sadig (msadig@uaeu.ac.ae)

Version: 4 Date: 22 May 2007

Author's response to reviews: see over
In response to formatting changes requested

- # 1  The test research article is removed
- # 2  The title is formatted as sentence case
- # 3  Author list is fixed
- # 4  Title and abstract are Fixed
- #5   Key words were removed
- #6   Competing interest is fixed
- #7   Figure is improved
- #8   Tables are inserted in the manuscript
- #9   Tables are listed in ascending order
- #10  Table 2 is fixed
- #11  References are fixed